Author's response to reviews

Title: Quality of health information for cervical cancer treatment on the internet

Authors:

Tara J Selman (taraselman@blueyonder.co.uk)
Trina Prakash (trina1811@aol.com)
Khalid S Khan (k.s.khan@bhan.ac.uk)

Version: 3 Date: 19 June 2006

Author's response to reviews:

Tara J Selman
Academic Department of Obstetrics and Gynaecology (3rd floor)
Birmingham Women's Hospital, Birmingham, B15 2TG
Phone: 07961053369
Fax: 0121 6272667
E-mail: taraselman@blueyonder.co.uk

The Editor

Dear Sir,

Re. Quality of health information for cervical cancer treatment on the internet

Thank you for your comments regarding the formatting of our study. We have made the recommended changes to the manuscript. The changes made are in the table following this letter.

Thank you once again for your consideration of this study.

Yours sincerely,

Tara J Selman

List of formatting changes made:

- Affiliations correct
- Authors qualifications removed
- References cited in text using consecutive numbers in square brackets
- Website addresses put in the reference list and removed from the text.
- Acknowledgement section added
- References changed to include all authors names
- Order of manuscript altered as requested
- Figure title removed from figure one and figure cropped